

We thank the reviewers for their comments. We have modified the manuscript as advised and the details are attached herewith.

	Original comments of the reviewer	REVISIONS DONE/REPLY TO PR COMMENTS
Reviewer #1:	<p>To review recent literature</p> <p>Improve figure 2</p> <p>Authors must homologate abbreviature</p> <p>Authors could consider representing data on relative risk, OR, HR, and IC</p> <p>Authors could improve ideas' order from general to particular because some sections are hard to read.</p>	<p>Recent literature was reviewed till June 2022 and new studies were included.</p> <p>Figure 2 was modified.</p> <p>Abbreviations was changed to be consistent across text. Data was modified as advised.</p> <p>Manuscript was modified as advised</p>
<i>Science editor</i>	<p>The abstract requires upgradation in alignment with the goal of the review</p> <p>Citations are inappropriately used</p>	<p>Abstract was modified as advised</p> <p>Citations was rechecked and edited accordingly.</p>